Author's response to reviews

Title: Association of 5HTR2A gene with suicidal behavior: Case-control study and updated meta-analysis.

Authors:

Thelma Gonzalez (thelma.glez.castro@gmail.com)
Carlos Tovilla (alfonso_tovillaz@yahoo.com.mx)
Isela Juarez (iselajua22@yahoo.com.mx)
Sherezada Pool (shepoga70@hotmail.com)
Martha Velazquez (patvel_7@hotmail.com)
Alma Genis (genis76@yahoo.com)
Humberto Nicolini (nicolini_humberto@yahoo.com)
Lilia Lopez (dralilialonar@yahoo.com.mx)

Version: 2 Date: 4 October 2012

Author's response to reviews: see over
Dear Deesha

We would like to submit to the consideration of your Editorial committee the paper entitled: “Association of 5HTR2A gene with suicidal behavior: Case-control study and updated meta-analysis” by Thelma Gonzalez, Carlos Tovilla, Juarez Isela, Alma Genis, Sherezada Pool, Humberto Nicolini and Lilia Lopez, as a Research Report in BMC Psychiatry.

The authors acknowledge that neither the manuscript nor its data have been previously published or are currently under consideration for publication elsewhere. Ethical approval of the study was granted and the research performed accordingly with the Code of Ethics of the World Medical Association (Declaration of Helsinki).

I declare that all authors participated in the design, analysis and interpretation of data in the paper and I also declare that I have no conflict of interest in connection with this paper other than any noted in the covering letter to the editor.

We think of our work as an interesting contribution to the field of genetic of suicidal behavior, to be considered for publication.

Sincerely,

Carlos Alfonso Tovilla, PhD.
Universidad Juárez Autónoma de Tabasco
División Académica Multidisciplinaria de Comalcalco
Ranchería Sur, Cuarta Sección, C. P. 86650
Comalcalco, Tabasco. México
Phone (52) 9933581500 ext 6900
E-mail: alfonso_tovillaz@yahoo.com.mx